



VOL. 7, 2024 A

ARTICLE 24581

ORIGINAL REPORT

FOLLOW-UP OF PATIENTS WITH POST COVID-19 CONDITION AFTER A MULTI-DISCIPLINARY TEAM ASSESSMENT: A PILOT STUDY

Alexander WIGGE, MD STUDENT¹, Johanna PHILIPSON, PhD², Solveig HÄLLGREN, MD¹, Helena FILIPSSON, RPT¹ and Britt-Marie STÅLNACKE, MD, PhD¹

From the ¹Department of Community Medicine and Rehabilitation, Rehabilitation Medicine, Umeå University, Umeå, Sweden, and ²Department of Clinical Sciences: Neurosciences, Umeå University, Umeå, Sweden

Objective: To follow up patients with post-COVID-19 condition (PCC) 6 months after a multidisciplinary team assessment in specialist care regarding symptoms of pain, anxiety, depression, fatigue and cognition, level of activity, physical activity and sick leave.

Methods: A prospective pilot study conducted in a clinical setting of patients (n = 22) with PCC referred from primary healthcare to a specialist clinic for a 2 day-multidisciplinary team assessment followed by a subsequent rehabilitation plan. Data were collected through questionnaires filled in prior to the team assessment and 6 months later. Results: Fifteen of the initial 22 patients participated in the follow-up. No statistically significant improvements were seen in any of the questionnaires after 6 months. However, 76.9% of the participants perceived the intervention as being helpful. This differed between the genders, where all the women 100% (n=8) perceived it as being helpful, compared with 40% (n=2) of the men (p=0.012). Conclusions: Based on these findings, the benefit of a multidisciplinary team assessment of PCC is not fully convincing. However, since the participants themselves perceived the intervention as being helpful, the team assessment seems to be of some value. Further studies with larger populations would be of interest.

Key words: post-COVID-19 condition; residual symptoms; rehabilitation.

Submitted Nov 6, 2023; Accepted Aug 22, 2024; Published XX.

Published Sep 23, 2024. DOI: 10.2340/jrmcc.v7.24581

JRM-CC 2024; 7: jrmcc24581

Corresponding address: Britt-Marie Stålnacke, Department of Community Medicine and Rehabilitation, Rehabilitation Medicine, Umeå University, Umeå, Sweden. E-mail; britt-marie.stalnacke@umu.se

LAY ABSTRACT

Most people who become ill with COVID-19 experience mild symptoms and recover within days or weeks. However, minor symptoms last from weeks to months and include fatigue, attention disorder, headache, dyspnea, anxiety, pain, anosmia, etc. This clinical state has been denoted as "post-COVID-19 condition" (PCC). It has been recommended that complex cases of PCC are to be assessed by a multidisciplinary team regarding treatment and rehabilitation. The efficiency of this recommendation has not yet been proven. This pilot study aimed to follow up patients with PCC six months after a multidisciplinary team assessment in specialist care regarding symptoms of pain, anxiety, depression, fatigue and cognition, level of activity, physical activity and sick leave through questionnaires. The results showed no statistically significant improvements in any of the questionnaires after six months. The majority of the participants perceived the team assessment as being helpful. Since the assessment seemed to be of some value, there is a need for further studies with larger populations.

A t the end of 2019, a newly discovered coronavirus, Severe acute respiratory syndrome corona virus 2 (SARS-CoV-2), was found to be the causative agent of the disease COVID-19. Most people who became ill with COVID-19 experienced mild symptoms and recovered within days or weeks. However, a minority became seriously ill with a need for hospitalization (1). Global mortality has been estimated to be 1% and in Sweden 0.8% (2, 3).

Although a majority of those infected by SARS-CoV-2 returned to their normal state of health once the infection had been cleared, many patients continued to

2024 ©Author(s). Published by MJS Publishing, on behalf of the Foundation for Rehabilitation Information. This is an Open Access article distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/).



p. 2 of 8 Follow-up of patients with post covid-19 condition

be symptomatic or developed new symptoms due to the infection. These sequelae can last for weeks to months, and include fatigue, attention disorder, headache, dyspnea, anxiety, pain, anosmia, ageusia etc. (4–6). This clinical state has been denoted as "post covid-19 condition" (PCC). According to WHO, PCC refers to long-term symptoms experienced after a confirmed or probable SARS CoV-2 infection. Individuals with PCC remain symptomatic or develop new symptoms within 3 months after falling ill. The symptoms persist for at least two months and cannot be explained by an alternative diagnosis (7).

Studies have shown that the prevalence of PCC varies. Some reports estimate that 3–10% have been suffering from symptoms of PCC for 12 weeks or more (8, 9). The severity of the acute disease has not been shown to affect the risk of acquiring PCC (10). However, it has been reported that the female sex, as well as increasing age and body mass index (BMI) are factors associated with increased risk of PCC (9).

The mechanisms of PCC are largely unknown. It has not been established whether parts of PCC are caused by the infection itself or if the condition is triggered by underlying chronic disorders. In addition, the effects of the pandemic on the individual have not been thoroughly investigated. The following are some of the hypotheses regarding the pathophysiology of PCC: harbouring of the virus in tissue reservoirs across the body; immune exhaustion with delayed viral clearance, hence causing chronic inflammation; and autoimmunity caused by cross-reactivity between SARS-CoV-2-specific antibodies and host proteins (11).

Since PCC can present a wide array of symptoms, the National Board of Health and Welfare in Sweden has recommended that complex cases of PCC are to be assessed by a multidisciplinary team regarding treatment and rehabilitation (12). The efficiency or efficacy of this recommendation has not yet been proven through clinical studies. Although a team assessment and rehabilitation should be based on a biopsychosocial approach, the length and intensity of a rehabilitation intervention for PCC has not been defined.

Therefore, this pilot study aimed to investigate the results of a 2-day multidisciplinary team assessment in a specialist care setting six months after the initial assessment with a focus on level of activity, sick leave, pain, anxiety, depression, fatigue and cognitive impairment. An additional aim was to study relationships between some of these variables.

MATERIAL AND METHODS

Design

A prospective cohort study.

Setting

The pilot study was conducted in a clinical setting in specialist healthcare between May 17 and December 22, 2021, with pauses of a total of 7 weeks during the summer and autumn.

Participants

Patients with PCS who were referred from primary healthcare centres in the County Council of Västerbotten to a multidisciplinary team at the neuro-head-neck centre (NHHC), Umeå University Hospital in Umeå, Sweden.

Inclusion criteria:

- Age at inclusion >18 years, with a history of probable or confirmed SARS CoV-2 infection assessed at a primary healthcare centre.
- Remaining symptoms of pain or neurological deficits such as fatigue or cognitive difficulties, for more than 12 weeks after falling ill. Symptoms of at least a moderate degree and assessed to be caused, or clearly worsened, by the SARS CoV-2 infection.
- Unimodal treatment with no obvious effect, whereby specialized multidisciplinary assessment regarding rehabilitation was deemed as being necessary.
- A total of 22 consecutive patients were included in the study.

Procedure

Data were gathered in the form of questionnaires filled out at baseline assessment and at follow-up after six months. Participants were contacted by telephone and reminded about the questionnaires. Following this, 15 of the initial 22 participants filled out the questionnaires, while seven participants withdrew from the study. Data from questionnaires were complemented by data from medical records.

Questionnaires

The Occupational Gap Questionnaire (OGQ) was used to measure participation in everyday activities (13). It consists of 30 items, each representing an activity. Every item is paired with two questions for the individual to answer. The questions are "Do you perform this activity?" and "Do you want to perform this activity?" Discrepancy between the two answers constitutes an occupational gap for that activity. Not want to do-gaps represent activities the individual does not want to perform but nevertheless does. Want to do-gaps are activities the individual wants to perform but nevertheless does not. These two kinds of gaps have been treated equally in the analysis, in accordance with previous literature (14). All items with incomplete or inexplicit answers in the OGQ were disregarded as missing data points.

The *Frändin Grimby activity scale* was used to assess level of physical activity (15). The questionnaire contains a 0–6 scale, with each number representing a degree of physical activity performed in everyday life. On this scale, 0 represents the lowest degree of physical activity and 6 the highest degree. Some participants marked two options on the scale. In these cases, the lower of the two values was chosen for analysis.

The Hospital Anxiety and Depression Scale (HADS) was used to assess anxiety and depression (16). It consists of 14 questions divided into two subscales: anxiety and depression. Each question has four alternatives with a score of 0–3 points. The total score range in both scales is 0–21. A score of 8–10 on one of the subscales suggests mild disorder, while a score >10 indicates there is a clinically significant disorder (16). The Mental Fatigue Scale (MFS) was used to assess symptoms of mental fatigue. It consists of 15 questions representing different symptoms of mental fatigue. Each question has seven alternative answers, where each alternative represents a certain

p. 3 of 8 Follow-up of patients with post covid-19 condition



number of points, ranging from no symptoms to severe symptoms. The cut-off score is 10.5 to distinguish between pathologic and non-pathologic mental fatigue. Scores exceeding 10 indicate that further investigation may be necessary. In this study, the MFS was only filled out at follow-up (17).

A multiple-choice questionnaire (MCQ) contained multiplechoice questions with 41 items. These questions focused on whether the individual experienced specific symptoms such as pain, fatigue, breathing difficulties or sleep disturbances. If they did, a quantifying alternative was to be chosen. There were also questions about work status and one question regarding whether they perceived the intervention with the 2-day multidisciplinary team assessment followed by a rehabilitation plan as being helpful. Nine items matching the aim of the study were chosen.

In the MCQ, answers in between options quantifying the given problem were counted as if the option representing the more serious problem had been chosen. All items with several answers, or items where the participant failed to choose a quantifying option, were disregarded as missing data points. The alternatives "minor problems" and "moderate problems" were grouped into the same category. For the drop-out analysis, the items were recoded into binary "yes" or "no" questions.

Multidisciplinary assessment

Patients were assessed by a multidisciplinary team, which consisted of a physician, specialist in rehabilitation medicine, a neuropsychologist, a physiotherapist, an occupational therapist and a counsellor. All professionals had long experience of multidisciplinary team assessment and rehabilitation. Each assessment lasted from 1 to 2 h. The team assessment was based on the biopsychosocial approach, which considered physical, psychological and social aspects of the PCC condition. The physician's assessment included a thorough medical examination with blood pressure and screening for orthostatic symptoms. The physiotherapist measured oxygen saturation at 1 min Sit-To-Stand Test and during 6-Min Walk Test. All the patients had normal levels of oxygen saturation during the tests. If there had been some questions and need for consultation a specific cardiologist was in service. Postural orthostatic tachycardia syndrome (POTS) was primarily investigated in primary care.

The individual assessment for each patient took place over the course of 2 consecutive days, during which the patients also filled out the questionnaires for the first time. At the end of day two, the team discussed their individual assessments to agree on a joint recommendation regarding rehabilitation. Afterward, the patients were invited to join the meeting where they were informed about the common assessment and recommendations made by the team. A rehabilitation plan was formed that included the team assessment of the patient's condition and suggestions and recommendations for further rehabilitation. The individual rehabilitation plan was sent to the patient's general practitioner to monitor.

Rehabilitation was then handled by the participants along with their respective primary healthcare centre for the following six months.

Rehabilitation plan

The rehabilitation recommendations were based on individual needs. Some of the recurring recommendations were:

- Pulse-raising physical activity, where effort was to be limited by the degree of symptoms. When a certain level of effort had been feasible for a certain amount of time, effort could be increased slightly.

- Usage of cognitive aids for planning, such as calendars and lists.
- Regular meetings with a counsellor or psychologist.
- With the help of an occupational therapist, find a balance between activities that were strenuous for the participant and recovery.
- Gradual increase of workload.
- Regular mini-breaks at work to avoid triggering severe fatigue.
- Avoid exposure to excessive sensory stimuli in order to conserve energy.
- Pharmaceutical treatment of insomnia, pain and depressive symptoms.

Data analysis

SPSS version 28.0.1.1 was used to process statistics. Nonparametric statistics were used for all the questionnaires. Descriptive statistics were used to describe the population. The Wilcoxon signed ranks test was used to compare results before assessment and at follow-up. Differences between the participants and the non-participants were analyzed using the Wilcoxon–Mann–Whitney test and the Chi-square test. Tests of correlation between variables were made with the Spearman correlation test. The *p*-value threshold was 0.05 throughout the study.

RESULTS

Drop-out analysis

At the time of multidisciplinary assessment, 22 patients participated in the study. By the time of the follow-up, seven had withdrawn from the study. There were no significant differences between participants and those who withdrew regarding age, gender, total HADS score and items from the MCQ.

Demographics

Demographics are presented in Table I. Of those who provided information about hospitalization during infection, hospitalization occurred in 33.3% (n=5). Three of the hospitalized patients provided information about BMI, all exceeding 25.

Previous cardiovascular disease (i.e. hypertonia, angina pectoris, atrial fibrillation, and pulmonary valve stenosis) was present in 33.3% (n=5) of the participants. Asthma occurred in 20% (n=3). A total of 26.7% (n=4) had previously been diagnosed with mental health disorders including clinical burnout, depression and anxiety disorders. A total of 6.7% (n=1) suffered from chronic pain.

Outcome at 6-month follow-up

At the 6-month follow-up, there were no statistically significant changes regarding the total amount of occupational gaps in the OGQ (see Table II). When the results for men and women were examined separately, women reported more gaps at baseline assessment and follow-up

p. 4 of 8 Follow-up of patients with post covid-19 condition



Table I.	Demographics	of participants	
----------	--------------	-----------------	--

Variable	n=15*	%	Median	IQR
Sex				
Men	6	40.0	-	-
Women	9	60.0	-	-
Age	-	-	45	10.0
Body mass index				
18.5-25	5	33.3	-	-
25-30	3	20.0	-	-
30-35	1	6.7	-	-
>40	1	6.7	-	-
Missing data	5	33.3	-	-
Education				
Primary school	0	0	-	-
Secondary school	7	46.7	-	-
University	8	53.3	-	-
Time between infection and assessment				
Days**	-	-	313	222.5
Hospitalized during infection				
Yes	5	33.3	-	-
No	7	46.7	-	-
Missing data	3	20.0	-	-

n = 15 was the total number of study participants. **Calculating the median number of days between infection and assessment, data from 12/15 participants were used.

than men. *Travelling* was the item with the highest frequency of gaps for all groups. Another top five item with respect to frequency was *helping and supporting others*. In relation to other items, this item increased in frequency at follow-up for every group (see Table III).

There were no statistically significant changes in physical activity at baseline assessment compared with follow-up on the Frändin Grimby scale (see Table II).

Results from the HADS showed no significant changes either between baseline and follow-up for the whole population or between men and women. At baseline, 23.1% (n=3) of the participants who answered (n=13-14) had a score resembling a clinically significant disorder of anxiety and depression, respectively. At follow-up, these proportions were 28.6% (n=4) for anxiety and 14.3% (n=2)for depression. The HADS scores were generally higher among men than women, as seen in Table II.

The median value of fatigue on the MFS at follow-up was 18.25 (IQR=7.75), which is above the cut-off value of 10.5 points. The median scores for men and women separately were also above 10.5 points, as shown in Table II.

An analysis of the separate items chosen from the MCQ, as shown in Table IV, indicated no significant

changes between baseline and follow-up for any of the items. Concerning the item fatigue, 100% of those who answered, both at baseline and follow-up, reported that they suffered problems of fatigue.

All of the patients who had reported on the item regarding cognition, had experienced such issues at baseline, while 93.3% had cognition issues at follow-up

Regarding the question whether patients perceived the 2-day team assessment followed by a rehabilitation plan as being helpful, 100% of the women who answered the question reported it was helpful, while the proportion of men who found it helpful was 40% (p=0.012).

A majority of the participants were on some degree of sick leave, but there was no statistically significant change between baseline assessment and follow-up.

Correlations

A statistically significant positive correlation was seen for the patients at follow-up (n=13-14) between fatigue scores on MFS and depression scores on HADS (r=0.716, p=0.004). No significant correlation was found between the scores on MFS and the anxiety scores on HADS.

There was a statistically significant negative correlation between the scores for physical activity and total number of occupational gaps for the patients (n=11–14) both at baseline (r=-0.761, p=0.004) and follow-up (r=-0.733, p=0.004). A significant negative correlation was found between age and total number of occupational gaps at baseline (r=-0.556, p=0.039), but not at follow-up.

DISCUSSION

The aim of the study was to follow up patients with PCC six months after a 2-day multidisciplinary team assessment in specialist care regarding level of activity, sick leave, pain, anxiety, depression, fatigue and cognitive issues. No statistically significant changes were seen on any of the measurement instruments when comparing baseline assessment and follow-up. Fatigue and cognitive impairment occurred widely both at assessment and at follow-up. Anxiety scores on the HADS resembled a mild anxiety disorder among men both at baseline and at follow-up. However, a majority of the participants

Table II. Comparison of data at baseline and follow-up

	inpunison	or due			Tonom	ар 									
		All par	ticipants				Wo	omen		_					
	Baseline, <i>n</i> = 13-14		Follow $n = 1$	v-up, 3-14	-	Baseline, n=8		Follow-up, n = 7-8		_	Baseline, n = 5-6		Follow-up, n=6		
Instrument	Median	IQR	Median	IQR	P	Median	IQR	Median	IQR	P	Median	IQR	Median	IQR	Р
HADS-tot	9.0	14.5	10	16.75	0.593	9.0	8.25	8.5	7.75	0.733	14.0	21.5	17.5	19	0.144
HADS-A	6.0	9.0	5.0	10.5	0.504	5.5	4.75	3.5	5.5	0.785	8.0	13	10	11.75	0.279
HADS-D	6.0	6.5	5.0	5.0	0.876	5.5	5.5	5.0	3.25	0.336	6.0	8.5	7.0	6.5	0.129
Frändin Grimby	2.0	2.0	3.0	2.0	0.257	2.0	1.25	2.0	2.0	0.102	3.0	3.0	3.0	2.25	0.317
GAPS-tot	10.5	8.25	9.5	11.75	0.400	13.0	8.75	11.0	7.0	0.610	6.5	7.0	1.5	12.75	0.528
MFS			18.25	7.75				19.0	8.88				16.75	10.37	

HADS-tot – Total score on the Hospital Anxiety and Depression scale. HADS-A – Anxiety score on the Hospital Anxiety and Depression scale. HADS-D – Depression score on the Hospital Anxiety and Depression scale. Frändin Grimby – Score of physical activity on a 0–6 scale. GAPS-tot – Total number of occupational gaps from the Occupational Gap Questionnaire. MFS – Total score of points on the Mental Fatigue Scale.

p. 5 of 8 Follow-up of patients with post covid-19 condition



Table III. Frequencies of occupational gaps for each individual item of the OGQ

	Num	ber of re	eported	gaps	Num	ber of re among	eported women	gaps	Number of reported gaps among men				
	Bas n = 1	Baseline $n = 11 - 14$		Follow-up $n = 11 - 14$		Baseline $n = 6-8$		Follow-up $n = 7 - 8$		Baseline $n = 4 - 6$		Follow-up $n = 4 - 6$	
Activity	п	%	п	%	п	%	п	%	п	%	п	%	
Shopping for groceries	9	75	6	43	5	71	4	50	4	80	2	33	
Cooking	4	33	5	38	3	43	4	50	1	20	1	20	
Doing the laundry	4	30	5	36	2	29	3	38	2	33	2	33	
Cleaning	5	36	6	50	4	50	5	71	1	17	1	20	
Doing light maintenance	6	43	6	43	5	63	3	38	1	17	3	50	
Doing extensive maintenance	7	54	6	46	6	75	5	71	1	20	1	17	
Managing personal finances	1	7	1	7	0	0	1	13	1	17	0	0	
Transporting oneself	6	43	4	29	4	50	3	38	2	33	1	17	
Shopping	4	29	4	29	3	38	2	25	1	17	2	33	
Participating/taking interest in sports/physical activity	8	62	7	50	6	86	6	75	2	33	1	17	
Participating in outdoor activities	6	43	6	43	4	50	4	50	2	33	2	33	
Having/practising hobbies	5	38	5	36	4	57	5	63	1	17	0	0	
Participating in cultural activities	6	46	5	38	5	63	4	50	1	20	1	20	
Watching TV/listening to the radio	2	14	0	0	2	25	0	0	0	0	0	0	
Reading newspapers	2	14	2	15	2	25	1	14	0	0	1	17	
Reading books/magazines	6	43	5	38	5	63	4	50	1	17	1	20	
Writing emails, letters, books, poems	4	31	3	23	4	57	1	13	0	0	2	40	
Playing games	5	42	5	38	4	57	4	50	1	20	1	20	
Using computer and smartphone	0	0	1	8	0	0	1	13	0	0	0	0	
Spending time with partner/children	0	0	0	0	0	0	0	0	0	0	0	0	
Keeping contact/spending time with relatives, friends, neighbors	2	18	1	9	2	33	1	14	0	0	0	0	
Helping and supporting others	9	64	8	62	6	75	6	75	3	50	2	40	
Participating in local clubs' activities	6	43	4	33	4	50	3	38	2	33	1	25	
Practising religion	1	8	0	0	1	13	0	0	0	0	0	0	
Visiting restaurants, cafés, pubs, dance clubs	8	66	6	50	5	71	4	57	3	60	2	40	
Travelling	11	79	9	69	7	88	6	75	4	67	3	60	
Working	5	36	6	46	4	50	4	50	1	17	2	40	
Studying	4	29	4	33	2	25	3	38	2	33	1	25	
Taking care of and raising children	1	8	0	0	1	14	0	0	0	0	0	0	
Volunteering	6	43	5	42	4	50	4	50	2	33	1	25	

OGQ: Occupational Gap Questionnaire.

Table IV. Comparison of data depicting job status and self-experienced issues from MCQ

	All participants						Won	nen			Men					
	Baseline, n = 11-12		Follow-up n = 13-15			Baseline, n=8		Follow-up n=8-9			Baseline $n = 3 - 4$		Follow-up $n = 5-6$		_	
Item	п	%	п	%	Р	п	%	п	%	Р	п	%	n	%	Р	
Problems with daily life activities	12	100	12	80	0.679	8	100	8	88.9	1.000	4	100	4	66.7	0.655	
Small-moderate	5	41.7	8	53.3		3	37.5	5	55.6		2	50	3	50		
Big	7	58.3	4	26.7		5	62.5	3	33.3		2	50	1	16.7		
Problems with pain and sensation	10	90.9	11	73.3	0.892	8	100	6	66.7	0.458	2	66.7	5	83.3	0.317	
Small-moderate	4	36.4	7	46.7		3	37.5	3	33.3		1	33.3	4	66.7		
Big	6	54.5	4	26.7		5	62.5	3	33.3		1	33.3	1	16.7		
Problems with cognition	12	100	14	93.3	0.414	8	100	8	88.9	0.564	4	100	6	100	0.564	
Small-moderate	3	25	8	53.3		2	25	5	55.6		1	25	3	50		
Big	9	75	6	40		6	75	3	33.3		3	75	3	50		
Problems with fatigue	12	100	15	100	0.564	8	100	9	100	1.000	4	100	6	100	0.317	
Small-moderate	5	41.7	5	33.3		3	37.5	3	33.3		2	50	2	33.3		
Big	7	58.3	10	66.7		5	62.5	6	66.7		2	50	4	66.7		
Perceived helpfulness of intervention			10	76.9				8	100				2	40		
Small-moderate			9	69.2				8	100				1	20		
Big			1	7.7				0	0				1	20		
Employed	11	91.7	11	78.6	0.317	8	100	7	87.5	0.564	3	75	4	66.7	0.317	
Full-time (100%)	7	58.3	6	42.9		4	50	3	37.5		3	75	3	50		
Part-time (25–75%)	4	33.3	5	35.7		4	50	4	50		0	0	1	16.7		
Sick leave	10	83.3	9	60	0.187	7	87.5	7	77.8	0.480	3	75	2	33.3	0.180	
Full-time (100%)	6	50	5	33.3		5	62.5	4	44.4		1	25	1	16.7		
Part-time (25–75%)	4	33.3	4	26.7		2	25	3	33.3		2	50	1	16.7		
Early retirement	2	16.7	2	13.3	1.000	1	12.5	1	11.1	1.000	1	25	1	16.7	1.000	
Full-time (100%)	1	8.3	1	6.7		0	0	0	0		1	25	1	16.7		
Part-time (25–75%)	1	8.3	1	6.7		1	12.5	1	11.1		0	0	0	0		
Sickness compensation	0	0	3	21.4	0.157	0	0	1	11.1	0.317	0	0	2	40	0.317	

MCQ: multiple-choice questionnaire.

p. 6 of 8 Follow-up of patients with post covid-19 condition



perceived the intervention with the team assessment followed by a rehabilitation plan as being helpful.

The severity of the acute state of infection varied among the participants, where some had needed hospitalized care but others had not. This is in line with previous studies, showing that severity of disease does not seem to affect the risk of acquiring PCC (10, 18).

The prevalence of depression among participants in our study was almost twice as high compared with the average life-time risk of depression shown in a cross-national study (19). Another study of patients with COVID-19 has shown that previous depression, anxiety, and usage of antidepressants were risk factors for acquiring persistent fatigue 10 weeks after the acute state of infection (20). It is uncertain whether the prevalence in our study was high because of a non-representative population, or whether it corresponds with the claim of previous mental health disorders constituting risk factors. However, similar results were found in a study by Poyraz et al. where 284 patients were assessed 50 days after acute SARS-CoV-2 infection (21). The mean scores for anxiety and depression, as well as the proportions of clinically significant anxiety and depression disorders, were quite close to the proportions shown in our study.

Observed HADS scores of depression, and especially anxiety, were generally higher among men than women. This finding is in contrast with most previous studies about the epidemiology of these conditions, where women have been shown to run a higher risk of developing depressive disorders as well as anxiety disorders (22). The anxiety scores among men in our study were similar to those reported in a study on chronic pain (23), indicating that the levels of anxiety might be comparable in PCC and chronic pain at least in men.

There was a positive relationship between the scores for depression and fatigue in our study. A previous study has suggested a comorbidity of depression and fatigue, with partly overlapping symptoms (24). The correlation seen in our study might be applicable in a clinical setting, by potentially improving problems of fatigue through the treatment of depression. It is also worth noting that the median value on MFS was higher than the cut-off 10.5, which indicates pathologic mental fatigue.

The high levels of fatigue and cognitive impairment in our study, as shown by the scores on MFS and the MCQ, were about three to four times higher than the results in a systematic review by Ceban et al. in patients 12 weeks after acute COVID-19 infection (25). However, these differences might be due to the fact that one of the inclusion criteria in our study was neurological symptoms, which could have affected the prevalence of both fatigue and cognitive impairment in our sample.

Regarding the total number of occupational gaps on the OGQ, there were no significant changes between baseline assessment and follow-up. However, there was an interesting change of frequencies when looking at *helping and supporting others*. At follow-up, this item increased

in frequency in relation to other items. At the same time, items such as *shopping for groceries* and *doing extensive maintenance* decreased in frequency in relation to other items at follow-up. These changes depicted a rise in the wish to help others, in relation to personal chores. This could indicate that the participants had an easier time carrying out personal chores at follow-up, and thereby an urge to spend more energy on helping others.

Physical activity, as measured by the Frändin Grimby scale, showed no statistically significant changes between baseline assessment and follow-up. In the literature, there are documented positive effects of physical activity for several disorders and conditions, for instance, mental health disorders and fatigue (26, 27). One could therefore speculate that increased physical activity might have led to more overall improvements in our population. However, the fatigue itself could be a reason why physical activity did not improve to begin with.

Despite the fact that no significant improvements were shown, a majority of the participants perceived the intervention with the 2 day-multidisciplinary team assessment followed by a subsequent rehabilitation plan as being helpful. Potential reasons for this seemingly contradictory result could be the feeling/experience of being looked out for and taken seriously. Even though our results do not indicate any significant improvements after the assessment, there might be other aspects of the rehabilitation interventions that provide helpful coping strategies, thus conveying the impression that the intervention was helpful.

There was a statistically significant difference between men and women, where a larger proportion of women perceived the intervention as being helpful. This kind of disparity between genders has been shown in multimodal rehabilitation of chronic pain, with larger improvements in women compared with men (28). In a previous quality study by Stenberg et al. (29) of neck and back patients it was shown that gender seemed to affect expectations and experiences of healthcare generally. In their study the men did not doubt that they were entitled to be helped, and took help from healthcare for granted while more women sought help for problems they had experienced for a long time but could no longer control. They expressed the attitude that others might have had a greater need of healthcare and doubted the value of their pain in the eves of others.

We could only speculate whether this could be a factor that has influenced the results of our study.

However, there is an increasing knowledge about the natural course of post-COVID-19 symptoms. In a recent study from Japan of the 1391 adult patients diagnosed with COVID-19 it was revealed that the prevalence of post-COVID-19 symptoms after three months were 47.6% and after 1 year 31% (30). A minor proportion (12.6%) experiences symptoms lasting for over three months that interfered with daily life, however, based on these findings this study was conducted.

p. 7 of 8 Follow-up of patients with post covid-19 condition

While assessing the quality of this study, several factors must be considered. Some strengths are that the multidisciplinary team consisted of experienced clinicians and that several of the questionnaires have been used in previous studies on similar groups of patients. Although a team assessment and rehabilitation should be based on a biopsychosocial approach, the optimal duration and intensity of a rehabilitation intervention for PCC has not been defined. In contrast to the short assessments that are commonly offered in ordinary healthcare the well-planned multidisciplinary assessment in this study was 2 days.

Some limitations to consider are the lack of power calculation, as well as the small size of the population following the exclusion of many of the referred patients. The small population in turn could be a reason for results that lack statistical significance. A final factor to be taken into account is that the compliance of the rehabilitation in the rehabilitation plan was unknown since this was dealt with by the patients themselves along with their general practitioner at the primary healthcare centres. However, since the referring general practitioners had experiences from patients with chronic pain and were used to the process of a 2-day assessment and rehabilitation plans it was possible to assume that the rehabilitation plans would be monitored. There was no control nor follow-up questions regarding compliance.

This study showed no statistically significant improvements in PCC following multidisciplinary team assessment and a rehabilitation plan. Nevertheless, a majority of the participants found the intervention helpful. However, more research on the subject is needed, preferably studies on larger populations, with control groups and compliance controls. In addition, qualitative studies may answer why women scored the assessment as helpful despite the absence of self-scored improvements.

ACKNOWLEDGMENTS

The study was approved by the Swedish Ethical Review Authority (dnr: 2017/483-31).

Written informed consent was obtained from all participants before inclusion.

The authors have no conflicts of interest to declare.

REFERENCES

- Ochani R, Asad A, Yasmin F, Shaikh S, Khalid H, Batra S, et al. COVID-19 pandemic: from origins to outcomes. A comprehensive review of viral pathogenesis, clinical manifestations, diagnostic evaluation, and management. Infez Med 2021; 29: 20–36.
- WHO. COVID-19 weekly epidemiological update. WHO; 2022, p. 106. [Cited 2022 October 2] Available from: www.who.int
- Folkhälsomyndigheten. Bekräftade fall av Covid-19 i Sverige. [cited 2022 Nov 1]. Available from: https:// experience.arcgis.com/experience/09f821667ce64bf7be6 f9f87457ed9aa.
- 4. Lopez-Leon S, Wegman-Ostrosky T, Perelman C, Sepulveda R, Rebolledo PA, Cuapio A, et al. More than 50 long-term



effects of COVID-19: a systematic review and meta-analysis. Sci Rep 2021; 11: 16144. https://doi.org/10.1038/ s41598-021-95565-8

- Willi S, Lüthold R, Hunt A, Hänggi NV, Sejdiu D, Scaff C, et al. COVID-19 sequelae in adults aged less than 50 years: a systematic review. Travel Med Infect Dis 2021; 40: 101995. https://doi.org/10.1016/j.tmaid.2021.101995
- Huang C, Huang L, Wang Y, Li X, Ren L, Gu X, et al. 6-month consequences of COVID-19 in patients discharged from hospital: a cohort study. Lancet 2021; 397: 220–232. https:// doi.org/10.1016/S0140-6736(20)32656-8
- Soriano JB, Murthy S, Marshall JC, Relan P, Diaz JV. A clinical case definition of post-COVID-19 condition by a Delphi consensus. Lancet Infect Dis 2022; 22: e102–e107. https:// doi.org/10.1016/S1473-3099(21)00703-9
- National Institute for Health and Care Excellence. Clinical guidelines. COVID-19 rapid guideline: managing the longterm effects of COVID-19. London: National Institute for Health and Care Excellence (NICE); 2020.
- Sudre CH, Murray B, Varsavsky T, Graham MS, Penfold RS, Bowyer RC, et al. Attributes and predictors of long COVID. Nat Med 2021; 27: 626–631. https://doi.org/10.1038/ s41591-021-01292-y
- Anaya JM, Rojas M, Salinas ML, Rodríguez Y, Roa G, Lozano M, et al. Post-COVID syndrome. A case series and comprehensive review. Autoimmun Rev 2021; 20: 102947. https:// doi.org/10.1016/j.autrev.2021.102947
- Alkodaymi MS, Omrani OA, Fawzy NA, Shaar BA, Almamlouk R, Riaz M, et al. Prevalence of post-acute COVID-19 syndrome symptoms at different follow-up periods: a systematic review and meta-analysis. Clin Microbiol Infect 2022; 28: 657–666. https://doi.org/10.1016/j.cmi.2022.01.014
- Socialstyrelsen (2021). Postcovid kvarstående eller sena symtom efter covid-19 Stöd till beslutsfattare och personal i hälsooch sjukvården (del 2) https://www.socialstyrelsen. se/globalassets/sharepoint-dokument/artikelkatalog/ ovrigt/2021-4-7351.pdf.
- Eriksson G, Tham K, Kottorp A. A cross-diagnostic validation of an instrument measuring participation in everyday occupations: the Occupational Gaps Questionnaire (OGQ). Scand J Occup Ther 2013; 20: 152–160. https://doi.org/ 10.3109/11038128.2012.749944
- Guidetti S, Utbult M, Kamwesiga J, Eriksson G. Perceived occupational gaps among the Ugandan general population – a pilot study. S Afr J Occup Ther 2019; 49: 17–23. https:// doi.org/10.17159/2310-3833/2019/vol49n3a4
- Grimby G, Frändin K. On the use of a six-level scale for physical activity. Scand J Med Sci Sports 2018; 28: 819–825. https://doi.org/10.1111/sms.12991
- Zigmond AS, Snaith RP. The hospital anxiety and depression scale. Acta Psychiatr Scand 1983; 67: 361–370. https:// doi.org/10.1111/j.1600-0447.1983.tb09716.x
- 17. Johansson BBK, Rönnbäck L. Evaluation of the Mental Fatigue Scale and its relation to cognitive and emotional functioning after traumatic brain injury or stroke. Int J Phys Med Rehabil 2013; 2: 1–7.
- Moreno-Pérez O, Merino E, Leon-Ramirez JM, Andres M, Ramos JM, Arenas-Jiménez J, et al. Post-acute COVID-19 syndrome. Incidence and risk factors: a Mediterranean cohort study. J Infect 2021; 82: 378–383. https://doi. org/10.1016/j.jinf.2021.01.004
- Bromet E, Andrade LH, Hwang I, Sampson NA, Alonso J, de Girolamo G, et al. Cross-national epidemiology of DSM-IV major depressive episode. BMC Med 2011; 9: 90. https:// doi.org/10.1186/1741-7015-9-90
- Townsend L, Dyer AH, Jones K, Dunne J, Mooney A, Gaffney F, et al. Persistent fatigue following SARS-CoV-2 infection is common and independent of severity of initial infection. PLoS One 2020; 15: e0240784. https://doi.org/10.1371/journal.pone.0240784
- Poyraz B, Poyraz CA, Olgun Y, Gürel Ö, Alkan S, Özdemir YE, et al. Psychiatric morbidity and protracted symptoms after COVID-19. Psychiatry Res 2021; 295: 113604. https://doi.

p. 8 of 8 Follow-up of patients with post covid-19 condition



org/10.1016/j.psychres.2020.113604

- 22. Gater R, Tansella M, Korten A, Tiemens BG, Mavreas VG, Olatawura MO. Sex differences in the prevalence and detection of depressive and anxiety disorders in general health care settings: report from the World Health Organization collaborative study on psychological problems in general health care. Arch Gen Psychiatry 1998; 55: 405–413. https://doi.org/10.1001/archpsyc.55.5.405
- 23. Gerdle B, Åkerblom S, Stålnacke BM, Brodda Jansen G, Enthoven P, Ernberg M, et al. The importance of emotional distress, cognitive behavioural factors and pain for life impact at baseline and for outcomes after rehabilitation – a SQRP study of more than 20,000 chronic pain patients. Scand J Pain 2019; 19: 693–711. https://doi.org/10.1515/ sjpain-2019-0016
- Corfield EC, Martin NG, Nyholt DR. Co-occurrence and symptomatology of fatigue and depression. Compr Psychiatry 2016; 71: 1–10. https://doi.org/10.1016/j. comppsych.2016.08.004
- Ceban F, Ling S, Lui LMW, Lee Y, Gill H, Teopiz KM, et al. Fatigue and cognitive impairment in post-COVID-19 syndrome: a systematic review and meta-analysis. Brain Behav Immun 2022; 101: 93–135. https://doi.org/10.1016/j. bbi.2021.12.020

- Pearce M, Garcia L, Abbas A, Strain T, Schuch FB, Golubic R, et al. Association between physical activity and risk of depression: a systematic review and meta-analysis. JAMA Psychiatry 2022; 79: 550–559. https://doi.org/10.1001/ jamapsychiatry.2022.0609
- 27. Razazian N, Kazeminia M, Moayedi H, Daneshkhah A, Shohaimi S, Mohammadi M, et al. The impact of physical exercise on the fatigue symptoms in patients with multiple sclerosis: a systematic review and meta-analysis. BMC Neurol 2020; 20: 93. https://doi.org/10.1186/ s12883-020-01654-y
- Pester BD, Crouch TB, Christon L, Rodes J, Wedin S, Kilpatrick R, et al. Gender differences in multidisciplinary pain rehabilitation: the mediating role of pain acceptance. J Context Behav Sci 2022; 23: 117–124. https://doi. org/10.1016/j.jcbs.2022.01.002
- Stenberg G, Fjellman-Wiklund A, Ahlgren C. "Getting confirmation": gender in expectations and experiences of healthcare for neck or back patients. J Rehabil Med 2012; 44: 163–171. https://doi.org/10.2340/16501977-0912
- Sugiyama A, Takafuta T, Sato T, Kitahara Y, Yoshinaga Y, Abe K, et al. Natural course of post-COVID symptoms in adults and children. Sci Rep 2024; 14: 3884. https://doi. org/10.1038/s41598-024-54397-y